Author's response to reviews

Title: A comparative study between mixed-type tumours from human salivary and canine mammary glands

Authors:

   Marisa CLS Genelhu (genelhumc@uol.com.br)
   Sergio V Cardoso (cardososv@hotmail.com)
   Helenice Gobbi (hgobbi@medicina.ufmg.br)
   Geovanni D Cassali (cassalig@icb.ufmg.br)

Version: 4 Date: 14 September 2007

Author's response to reviews: see over
Minor Essential Revisions

- We standardized the form of indicating the statistical level of significance according to the reviewer's suggestion.

- We also corrected the typography of first author in references, the abbreviation RNAm for mRNA and others words were standardized (E.g.: malignant, antigenic, tumours…)

Discretionary Revisions

- In spite of the article suggested by the reviewer (In der Maur et al, 2005) to be more recent and to contain important information, we preferred to maintain the reference Wick et al, 1998. We consider it more appropriate and specific for ductal carcinomas.